Author's response to reviews

Title: Factors associated with pastoral community knowledge and occurrence of mycobacterial infections in Human-Animal Interface areas of Nakasongola and Mubende districts, Uganda

Authors:

Clovice Kankya (clokankya@yahoo.com)
Adrian Muwonge (adrian.muwonge@nvh.no)
Susan Olet (susan.olet@ndsu.edu)
Musso Munyeme (munyeme@yahoo.co.uk)
Demelash Biffa (Aredares@veths.no)
John Prof Opuda-Asibo (opas_in@yahoo.com)
Eystein Prof Skjerve (Eystein.Skjerve@nvh.no)
James As. Prof Oloya (joloya2001@yahoo.co.uk)

Version: 6 Date: 30 July 2010

Author's response to reviews:

To be sent later